Author’s response to reviews

Title: Interventions aimed at increasing the level of physical activity by including organised follow-up: a systematic review of effect.

Authors:

Eva Denison (Eva.Denison@nokc.no)
Gunn E Vist (GunnElisabeth.Vist@nokc.no)
Vigdis Underland (Vigdis.Underland@nokc.no)
Rigmor C Berg (Rigmor.Berg@nokc.no)

Version: 4 Date: 4 June 2014

Author's response to reviews:

Revisions are described in the document Response to Dr Kousoulis.